Author's response to reviews

Title: Human pericardium graft in the management of bleb's complication performed in childhood: a case report

Authors:

Dimitris Papaconstantinou (dpapaconstantinou@hotmail.com)
Ilias Georgalas (igeorgalas@yahoo.com)
Sergios Taliantzis (sergiotali@hotmail.com)
Chrysanthi Koutsandrea (ancode@otenet.gr)
Ioannis Ladas (idladas@gmail.com)
Gerassimos Georgopoulos (glaucomanet@aim.com)

Version: 5 Date: 23 July 2011

Author's response to reviews: see over
Cover letter

Dear editor

Reviewer Nitin Anand

We include a paragraph in the conclusions stating the advantages of human pericardium over donor scleral graft.

Reviewer Kazuyuki Hirooka

BCVA was still of light perception five months postoperatively because of end state glaucomatous optic disc atrophy.

The hypotony (IOP < 0 mmHg) was related to bleb leakage. Seidel test at presentation was negative because the choroidal tissue was incarcerated to the sclerectomy.

Declaration of competing interests was included, such as authors’ contributions

Changes in the manuscript are coloured and underlined

Sincerely yours

Taliantzis Sergios